Author's response to reviews

Title: Yield of Endoscopic ultrasound in patients with moderate and high likelihood of suspected choledocholithiasis: a retrospective study from a single university based- endoscopic center

Authors:

Varayu Prachayakul (kaiyjr@gmail.com)
Pitulak Aswakul (apitulak@gmail.com)
Patommatat Bhunthumkomol (asawakul@gmail.com)
Morakod Deesomsak (mkd2181@gmail.com)

Version: 2 Date: 22 April 2014

Author's response to reviews:

This was the revision version regarded the recommendation from the editor:
Point-by-point revision
1. We did insert all the authors’ email address in the title page
2. We did insert the authors' contribution section